Author's response to reviews

Title: Image-guided versus blind corticosteroid injections in adults with shoulder pain: A systematic review

Authors:

Edmund Soh (esoh@doctors.org.uk)
Wenyun Li (wenyun.li@scri.edu.sg)
Keh Oon Ong (ong.keh.oon@sgh.com.sg)
Wen Chen (chenwen@hitachi-aloka.com.sg)
Dianne Bautista (dianne.bautista@scri.edu.sg)

Version: 9 Date: 24 June 2011

Author's response to reviews: see over
Dear BMC Series Editorial Production Team,

Thank you for your email of 23rd June 2011.

Following on from this note is a summary of our responses to the editorial requirements. I have also uploaded separately the revised manuscript with highlighted tracked changes.

Yours faithfully,

Edmund Soh
esoh@doctors.org.uk
Summary of responses to editorial requirements

Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution. Also please link affiliation 4 to the author list

RESPONSE: Each affiliation has been written in the above format (we have assumed that street and postal code are optional). Please note that Singapore is both a city and country. Affiliation 4 has been linked to Wen Chen (it was already linked in the original manuscript).

Tables: Please submit table 1 as a figure file, adjust citations accordingly

RESPONSE: Table 1 has been resubmitted as Figure 2. The original Figures 2, 3 and 4 have been relabelled as Figures 4, 3 and 5 respectively; the swap of the original Figures 2 and 3 to Figures 4 and 3 is intentional to better fit the main manuscript. Citations in the main manuscript have been adjusted accordingly.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com

RESPONSE: The email addresses have been rewritten in the above format

Previous exhibition: If you have previously exhibited your work at a conference (or similar), details of this should be provided in the Acknowledgements only. Please remove this statement from the title page

RESPONSE: The conference statement has been removed from the title page. We will leave it out from the Manuscript.